Author's response to reviews

Title: Trigeminal neuralgia post-styloidectomy in Eagle syndrome: a case report

Authors:

  John W Blackett (jblackett@wusm.wustl.edu)
  Daniel J Ferraro (dferraro@radonc.wustl.edu)
  John J Stephens (stephensj@mir.wustl.edu)
  Joshua L Dowling (d BowlingJ@wudosis.wustl.edu)
  Jerry J Jaboin (jjaboin@radonc.wustl.edu)

Version: 2 Date: 20 August 2012

Author's response to reviews: see over
August 19, 2012

Dear Editors:

We are writing to respond to the reviews regarding our manuscript “Trigeminal neuralgia post-styloidectomy in Eagle syndrome: a case report.” We appreciate the time and thoughtful reviews provided from the editorial staff and peer reviewers. Below, we address each comment that was submitted and we also have included a revised manuscript in Microsoft Word format using the track changes function to easily identify changes made to the manuscript.

We hope you will find these revisions to your satisfaction and look forward to working with you to publish this in your journal in the near future.

Sincerely,

Jerry J. Jaboin, MD, PhD
Assistant Professor
Department of Radiation Oncology
Mallinckrodt Institute of Radiology
Washington University School of Medicine

Editor’s Comments:

In addition to the reviewer’s comments, please also address the following formatting changes required for your paper:
1. Please remove the dates found in the Case presentation section
2. Please remove any text from figure one, which may reveal the patient’s identity (e.g. patient’s number, patient’s name, dates, etc)

Changes have been made to the text and figure as requested.

Reviewer: Narayanan Prepageran
No changes requested.

Reviewer: Lum Chee Lun
In page 10, Line 5 and 11 - please substitute the word ’SCA’ to ’Superior Cerebellar Artery’.

The change was made to the manuscript as suggested.

Reviewer: Kin Choo Pua
No changes requested.